Author's response to reviews

Title: The relationship between mentoring on healthy behaviors and well-being among Israeli youth in boarding schools: A mixed-methods study

Authors:

Maayan Agmon (agmon.mn@gmail.com)
Cheryl Zlotnick (czlotnick@univ.haifa.ac.il)
Anat Finkelstein (fanat@zahav.net.il)

Version: 2 Date: 24 July 2014

Author's response to reviews:

July 24, 2014

Dear Editor:

Thank you for your prompt and detailed response.

According to your requests, we have made the following changes to the manuscript:

1. A declaration of no competing interest was added to the manuscript, placed after the conclusions.
2. Acknowledgements were added to the manuscript, after the authors’ contributions section, before the references.
3. Figure numbers and titles were added after the references, before the tables. No further descriptions are required for the submitted figures.

Sincerely yours,
Maayan Agmon.